\documentclass[12pt]{article}

\setlength{\parindent}{0pt}
\everymath={\displaystyle}
\pagestyle{empty}

\begin{document}

\textbf{Math 225: Calculus III}
\hfill\makebox[2.5in][r]{Name:\hrulefill}

\textbf{Quiz 5}\quad \textsl{October 10, 2002}
\hfill\makebox[2.5in][r]{Section:\hrulefill}

\bigskip

1. Find the local maximum and minimum values and saddle point(s)
of the function $f(x,y) = (x^2 + y^2)^2 + x^2 - 2y^2$.

\vspace*{3in}

2.  Find the absolute maximum value of the function $f(x,y) = x y^2 -
x^3$ on the square $0 \le x \le 1$, $0 \le y \le 1$.

\end{document}
